Author’s response to reviews

Title: An individualized patient-reported outcome measure (PROM) based patient decision aid and surgeon report for patients considering total knee arthroplasty: protocol for a pragmatic randomized controlled trial

Authors:

Nick Bansback (nick.bansback@ubc.ca)
Logan Trenaman (trenaman@alumni.ubc.ca)
Karen MacDonald (karenv.macdonald@ucalgary.ca)
Gillian Hawker (g.hawker@utoronto.ca)
Jeffrey Johnson (jeff.johnson@ualberta.ca)
Dawn Stacey (dstacey@uottawa.ca)
Deborah Marshall (damarsha@ucalgary.ca)

Version: 2 Date: 24 Jan 2019

Author’s response to reviews:

January 24, 2019

Dear Bethan Wicker,

Re: Revisions for Manuscript BMSD-D-18-00881R1

We would like to thank you on behalf of all the authors for the thoughtful comments on the manuscript. We have provided a response to the Editor’s comments.

If you have any questions or comments, please be in touch. We look forward to hearing from you.
Sincerely,

Nick Bansback
Associate Professor
School of Population and Public Health
University of British Columbia

Logan Trenaman
Graduate Student
School of Population and Public Health
University of British Columbia

Editor Comments

1. Please provide figure titles/legends under a separate heading of 'Figure Legends' after the References. If Figure titles/legends are within the main text of the manuscript, please move them.

Response: Please accept our apologies, the Figure Legends page was accidentally removed from the revised manuscript that was submitted during the revision. We have added the Figures Legends page after the References.

2. Thank you for providing your SPIRIT checklist. However, at this stage it is not required and so we kindly ask that you remove it from your manuscript.

Response: The SPIRIT checklist has been removed from the manuscript.
3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: A clean version of the manuscript has been uploaded with separate Figure files.